Author's response to reviews

**Title:** Cryptosporidiosis and Isosporiasis among HIV-positive individuals in South Ethiopia: a cross sectional study

**Authors:**

Mekonnen G Asfaw (mekbiomed2006@yahoo.com)
Wondu T Amenu (wondue_teshome@yahoo.com)
Beyene P Lodamo (abule2002@yahoo.com)
Tekola E Kassa (teko1960@yahoo.ca)

**Version:** 6  **Date:** 3 February 2014

**Author's response to reviews:** see over
Responses

First we would like to thank all the editors for their time to give us constructive comments and suggestions to improve this manuscript. We have, to our best capacity, tried to address the concerns:

1. Edit typos after some headings

   We have discovered that we have erroneously used colons after some headings in the methodology part and we have removes them

2. Change the term ‘predictors’

   Comment well taken and we have removed the term ‘predictors’ from headings of tables and in the text and replaced it with ‘associated factors’.

3. ‘et-al’… This is now rewritten as ‘et al’